Author's response to reviews

Title: Health-related Quality of life in 640 head and neck cancer survivors after radiotherapy using EORTC QLQ-C30 and QLQ-H&N35 questionnaires

Authors:

Stephen Wan Leung (alfrequent@gmail.com)
Tsair-Fwu Lee (tflee@cc.kuas.edu.tw)
Chih-Yen Chien (cychien3965@adm.cgmh.org.tw)
PJ Chao (pichao99@gmail.com)
Wen-Ling Tsai (a7323630@ms14.hinet.net)
Fu-Min Fang (fang2569@adm.cgmh.org.tw)

Version: 7 Date: 4 April 2011

Author's response to reviews: see over
Dear Editor,

All the words corrected were written in blue color in the revised manuscript.

Best regards,

Tsair-Fwu Lee